Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
April 14, 2010
Injector    Energy 56.62 MeV
· Injector needs optimization before Hall A is back. After yesterday’s beam studies the injector seems different.

· QE measurements did not work. It gave zero QE. It is working now. 

· (4/10)The GUN HV set point was changed somehow from 100.15 to 100. It is possible that something is wrong with the software. If there are the GUN comparator FSD trips check the GUN HV setpoint.
· Injector group has been searching for the cause of the beam motion (0.2 Hz) in the injector starting 1I04 but has not found anything yet. The motion is very small (<0.1mm).
· .

· A spin dance was performed to optimize the beam polarization. As a result of the spin dance the Wien angle was changed from -59 to -52 degrees. After the spin dance Hall A took beam with vertical polarization during Tuesday owl shift. The vertical polarization was backed out during Tuesday day shift.
· (4/2) Iocin1mag crashed. After the ioc was back beam would not reproduce in the injector.

· Sometimes the laser phases need to be adjusted a few degrees for reasons unknown.

Accelerator   Linac Energy 503.25 MeV
· The accelerator was running well with low current in the machine.
· Last night swing shift after beam studies / maintenance the witness beam to the BSY could not be restored. We need check the machine (possibly sometime today).

· RF 2L07 phantom trips returned. After yesterday’s troubleshooting the zone was bypassed. 

· MQA8A25 voltage is not stable, which needs further investigation. MQA1H04A may have similar problems too (currently it is set to zero field).
· RF 1L06 had wrong Q and the problem was corrected.

· The viewer lights were fixed for ITV6E02 &ITV8E02

· The accelerator has been running well with low trip rates.

· There were unusual BLM activities at the end of NL. It was caused by Hall C beam. Hall C laser phase was adjusted and the BLM activities are back to normal.

· (4/6) 2L07 phantom P1 fault was resolved by replacing the FSD card.
· (4/2) The air hose to VBV1H00 was loose and fixed.

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· One of the Hall A targets was damaged most likely during April 6 Owl shift beam delivery. See Hall A log entry # 316718 for details. Also the vacuum seal O-ring downstream the target was damaged from radiation, which caused the vacuum leak.
Hall A has been down due to its target and vacuum issues. Hall A has not given Ops an official report yet regarding the target incident. Hall A changed the current limit on the damaged target. Hall A is putting a Tungsten collimator into the beampipe to protect the vacuum seal.
· Prior to resuming beam delivery to Hall A we need to do the ion chamber functional test and reset ion chamber limits.

· One of the Hall A targets was possibly damaged between 0045 and 0830 yesterday’s owl shift. The final report is not out yet. The hall has been down since 0830 yesterday.
· Hall A line was matched and the beam size is small now (about 60 um x 60 um). Hall A Compton rates were lowered by closing the hall slit. Without closing slit the Compton rates are still high, which means that something in the injector affects the Compton rates.

· Hall A Moller commissioning went well so far. They have not tried Moller with the special beam structure yet.

Hall B ---Pass 3, g9-Frost from March 11 to July 22, 2010
· No problems with Hall B beam

· Hall B beam is fine.
· Last night swing shift after the Hall was back online the beam was not good-too noisy. An early allsave was downloaded to correct the problem.

· Hall B had beam off for four shifts (Monday and Tuesday) for thermal equilibrium of its target.

Hall C---QWeak from May 25 to July 22, 2010
· QWeak installation
